Author’s response to reviews

Title: Contribution of main causes of death to social inequalities in mortality in the whole population of Scania, Sweden

Authors:

Maria Rosvall (Maria.rosvall@med.lu.se)  
Basile Chaix (Basile.chaix@med.lu.se)  
John Lynch (John.lynch@mcgill.ca)  
Martin Lindstrom (Martin.lindstrom@med.lu.se)  
Juan Merlo (Juan.merlo@med.lu.se)

Version: 5 Date: 13 March 2006

Author’s response to reviews:

The Editor, March, 2006
BioMed Central

Dear Sir/Madame:

Thank you for accepting our manuscript "Contribution of main causes of death to social inequalities in mortality in the whole population of Scania, Sweden" (MS: 1993632551890351)" for publication in BMC Public Health. We have now made the format changes to the manuscript according to the instructions sent to us by e-mail. Here is a description of the changes made.

All correspondence concerning this manuscript may be addressed to:
Maria Rosvall, MD, PhD  
Department of Community Medicine  
University Hospital, Malmo  
SE-205 02 Malmo, Sweden.  
Tel + 46, 40 33 10 00  
Fax+ 46, 40 33 70 96  
E-mail: maria.rosvall@smi.mas.lu.se

Sincerely,  
Maria Rosvall

Format changes:  
The title page now includes the manuscript title and the full names and affiliations for all authors. The corresponding author is indicated. The title page is now included in the manuscript.  
Abstract- The capitals have been taken out and the abstract has been shortened to 348 words.  
Key words - The key words have been taken out from the manuscript.  
References - The references are now cited in square brackets.  
Figures - The figures are cropped as closely as possible.
Additional files - The additional files with the title page and cover letter have been deleted.